Author's response to reviews

Title: Knowledge translation strategies to improve the use of evidence in public health decision-making in local government: a logic model for intervention design.

Authors:

Rebecca Armstrong (armr@unimelb.edu.au)
Elizabeth Waters (ewaters@unimelb.edu.au)
Maureen Dobbins (dobbinsm@mcmaster.ca)
Laurie Anderson (lmander@uw.edu)
Laurence Moore (MooreL1@cf.ac.uk)
Mark Petticrew (Mark.Petticrew@lshtm.ac.uk)
Rachel Clark (rachelclark@ceips.org.au)
Tahna Pettman (tpettman@unimelb.edu.au)
Catherine Burns (cate.m.burns@gmail.com)
Marj Moodie (marj.moodie@deakin.edu.au)
Rebecca Conning (Rebecca.conning@gmail.com)
Boyd Swinburn (boyd.swinburn@deakin.edu.au)

Version: 4 Date: 18 September 2013

Author's response to reviews:

Thank you for the editorial comments.
1. I have marked the corresponding author.
2. Catherine Burns does not have a formal affiliation at the moment.
3. First names of authors provided.
4. References reformatted.
5. All other changes are fine and I have accepted them.
Kind regards,
Rebecca